Author's response to reviews

Title: 'It is like a tomato stall where someone can pick what he likes': Structure and practices of female sex work in Kampala, Uganda.

Authors:

Sarah Nakamanya (sarah.nakamanya@mrcuganda.org)
Martin Mbonye (martin.mbonye@mrcuganda.org)
Winfred Nalukenge (winfred.nalukenge@mrcuganda.org)
Rachel King (rlking@psg.ucsf.edu)
Judith Vandepitte (judith.vandepitte@mrcuganda.org)
Janet Seeley (janet.seeley@mrcuganda.org)

Version: 2 Date: 27 June 2013

Author's response to reviews:

Letter attached.